Author's response to reviews

Title: Combined maxillomandibular swing approach and neurosurgical ultrasonic aspirator for extensive Clival chordoma: a case presentation.

Authors:

Shahid Hassan (shahid@kb.usm.my)
Jafri M Abdullah (neurosains@kck.usm.my)
Shah Jihan Wan Din (wsjihan@kb.usm.my)
Zamzuri Idris (zamzuri@kb.usm.my)

Version: 2 Date: 29 August 2007

Author's response to reviews:

Dear Editor,

The changes have been made as per request.

1. subheading changed to case presentation
2. patient initial removed
3. word count, key word and few subheading removed
4. abbreviation and acknowledgment added
5. first two figure removed and legends have been put at appropriate place.

thank you.

Yours truely,

Asso Prof Dr Shahid Hassan